Author's response to reviews

Title: Prognostic value of p27Kip1 expression in Basaloid Squamous Cell Carcinoma of the larynx.

Authors:

Grazia Salerno (grazia.salerno@tin.it)
Dolores Di Vizio (Dolores.DiVizio@chidrens.harvard.edu)
Stefania Staibano (stайлano@unina.it)
Giampiero Mottola (giampieromottola@hotmail.com)
Giuseppe Quaremba (guaremba@unina.it)
Massimo Mascolo (mmascol@libero.it)
Vieri Galli (vieri.galli@unina.it)
Gaetano De Rosa (gaderosa@unina.it)
Luigi Insabato (insabato@unina.it)

Version: 7 Date: 31 May 2006

Author's response to reviews:

Title: Prognostic value of p27Kip1 expression in Basaloid Squamous Cell Carcinoma of the larynx.
Version Date: 31 May 2006

COVER LETTER

The requested formatting changes were done:

- Affiliations (country always included)
- Section heading (unnecessary capitals were removed)
- Tables (all the tables were renamed using standard numbers and the references to them throughout the manuscript text were changed)